Author's response to reviews

Title: HIV and pre-neoplastic and neoplastic lesions of the cervix in South Africa: a case-control study

Authors:

Jennifer R Moodley (moodley@cormack.uct.ac.za)
Margaret Hoffman (mh@cormack.uct.ac.za)
Henri Carrara (carrara@ukzn.ac.za)
Bruce R Allan (ballan@curie.uct.ac.za)
Diane D Cooper (dic@cormack.uct.ac.za)
Lynn Rosenberg (lrosenberg@slone.bu.edu)
Lynette E Denny (ldenny@uctgsh1.uct.ac.za)
Samuel Shapiro (shapiro@cormack.uct.ac.za)
Anna-Lise Williamson (annalise@curie.uct.ac.za)

Version: 4 Date: 18 May 2006

Author's response to reviews: see over
Re: Formatting changes MS 196234784932327

On behalf of my co-authors, I submit the revised manuscript “The relationship between HIV and pre-neoplastic and neoplastic lesions of the cervix in South Africa: a case-control study.” All formatting changes requested by the Editorial Team have been made.

Yours sincerely

Dr J Moodley